Author's response to reviews

Title: Changes in weight loss, body composition and cardiovascular disease risk after altering macronutrient distributions during a regular exercise program in obese women

Authors:

Chad M Kerksick (Chad_Kerksick@ou.edu)
Jennifer Wismann-Bush (Jen_Wismann@baylor.edu)
Donovan Fogt (Donovan.Fogt@utsa.edu)
Ashli R Thomas (huskertart@yahoo.com)
Lemuel Taylor IV (ltaylor@umhb.edu)
Bill Campbell (Campbell@coedu.usf.edu)
Colin D Wilborn (cwilborn@umhb.edu)
Travis Harvey (travis.harvey@usma.edu)
Mike Roberts (Mike_Roberts@ou.edu)
Paul La Bounty (Paul_La_Bounty@baylor.edu)
Melynn Galbreath (Maly_Galbreath@baylor.edu)
Brandon Marcello (bmarcello@stanford.edu)
Chris Rasmussen (crasmussen@hlkn.tamu.edu)
Richard B Kreider (rkreider@hlkn.tamu.edu)

Version: 2 Date: 3 May 2010

Author's response to reviews:

A statement regarding IRB approval and compliance with Declaration of Helsinki has been added and all extra spaces have been removed throughout document.